Author's response to reviews

Title: Correlations between prescription of anti-hypertensive medication and mortality due to stroke

Authors:

Renata Papp (papp@oali.hu)
Albert Csaszar (drcaszalb@gmail.com)
Edit Paulik (paulik@puhe.szote.u-szeged.hu)
Sándor Balogh (balogh@oali.hu)

Version: 3 Date: 9 February 2012

Author's response to reviews:

For attn of Marinette Lacson
Journal Editorial Office
BioMed Central
Dear Ms Lacson,

Thank you for considering our article for publication in BMC Cardiovascular Disorders.
We have adjusted the section of "Conclusions" to the comments received. We have marked the italic.
We emphasized that the conclusions are limited to the three counties examined. To be generalized to the whole country would have limitations. The findings could show us trends but not sharp conclusions.
We have uploaded the changes on the site.
Hope the answers provide enough details to fulfill expectations of the Editorial Board.
Looking forward to a positive decision,

Kind regards,

Albert Császár